Author's response to reviews

**Title:** A rare case of term viable secondary abdominal pregnancy following rupture of rudimentary horn: a case report.

**Authors:**

Bhandary Amritha (amrithabhandary@rediffmail.com)
Thirunavukkarasu Sumangali (dsumangali_vijay@yahoo.co.in)
Ballal Priya (priyab@yahoo.co.in)
Shedde Deepak (deepaks@yahoo.com)
Rai Sharadha (sharadar@rediffmail.com)

**Version:** 2  **Date:** 17 February 2008

**Author's response to reviews:**

TO,

The editor,

Journal of medical case reports.

Sir,

As per the required changes mentioned in the mail sent to me, i have made the changes.

The title has been mentioned as ending with "a case report" and the title page numbered as pager 1. The abstract containing background, case in brief and conclusion has been added to page 2. Consent has been modified and the figure legend has been included. sorry for the delay in reply. kindly consider my case report for review. Looking forward for your reply.